Author’s response to reviews

Title: Pulmonary Tuberculosis among Women Attending Family Planning and Maternal and Child Health Clinics in Dar Es Salaam, Tanzania. AUTHORS: Esther S Ngadaya1 2* $, Godfrey S Mfinanga2*, Eliud R Wandwalo3* and Odd Morkve1*. 1 Centre for International Health, University of Bergen, Norway. 2 National Institute for Medical Research, Muhimbili Medical Research Centre, P.O. Box 3436, Dar es Salaam, Tanzania 3 Ministry of Health and Social Welfare, National Tuberculosis and Leprosy Control Programme, (NTLP), P.O.Box 9083 Dar es Salaam, Tanzania. *These authors contributed equally to this work $Corresponding author Email addresses: ESN: engadaya@yahoo.com GSM: gsmfinanga@yahoo.com ERW: ewandwalo@hotmail.com OM: odd.morkve@cih.uib.no

Authors:

Esther S Ngadaya (engadaya@yahoo.com)
Godfrey S Mfinanga (gsmfinanga@yahoo.com)
Eliud R Wandwalo (ewandwalo@hotmail.com)
Odd Morkve (odd.morkve@cih.uib.no)

Version: 2 Date: 23 January 2009

Author’s response to reviews:

No revisions just included adresses of the authors